Author's response to reviews

Title: A multi-disciplinary education process related to the discharging of children from hospital when the child has been diagnosed with type 1 diabetes - a qualitative study

Authors:

Lisbeth Jönsson (lisbeth.jonsson@med.lu.se)
Inger Hallström (inger.hallstrom@med.lu.se)
Anita Lundqvist (anita.lundqvist@med.lu.se)

Version: 8 Date: 14 May 2010

Author's response to reviews:

Not relevant